Author's response to reviews

Title: Coronary bypass grafting using Crossclamp Fibrillation does not result in reliable reperfusion of the Myocardium when the crossclamp is intermittently released: A prospective cohort study

Authors:

Joel Dunning (joeldunning@doctors.org.uk)
Steven Hunter (Steve.Hunter@stees.nhs.uk)
Simon W.H. Kendall (Simon.Kendall@stees.nhs.uk)
John Wallis (John.Wallis@stees.nhs.uk)
William A Owens (Andrew.Owens@stees.nhs.uk)

Version: 3 Date: 21 November 2006

Author's response to reviews:

Dear Editor

Thank you for accepting the paper. We have changed the author list, deleted the title, author title, word count, keywords, moved the references, and deleted the figures out of the text and removed the highlighting.

I look forward to the final version

Yours sincerely

Joel Dunning